%  File: all_brec.tex              20-NOV-1991

%  Bibliographic entries for all of the papers in the Procedings
%
%  Run:  latex all_brec
%        bibtex all_brec
%        latex all_brec
%        latex all_brec
%  Requires:  twoside.sty, siam11.sty, breck_hd.tex
%             all_brec.bib, brecsiam.bst
%     brecsiam.bst is the same as siam.bst except that the items
%     are printed with keys of the form [Auth99a] rather than [99].

\documentstyle[twoside,siam11]{article}
\input {breck_hd}

\markboth {Part IX.  Bibliography}
          {Automatic Differentiation Bibliography}


\title {Automatic Differentiation Bibliography}
\author {collected by George F. Corliss}
\date {}


% Commands to get Russian digraphs:
\newcommand{\Yu}{{Yu}}
\newcommand{\Ye}{{Ye}}
\newcommand{\Ju}{{Ju}}


\begin {document}
\maketitle
\onelinetitle

    This bibliography represents the common bibliography for all of the
papers in this volume.  Each author prepared a bibliography for her or
his own paper.  The separate bibliographies were merged into a single
BibTex database, and references from several other sources were added.
Especially valuable contributions were made by bibliographic data
bases previously compiled by Bruce Char, by David Gay and by
Davis, Corliss, and Krenz [Davi88a].

    This bibliography includes most of the work known to the editors 
in the area of automatic differentiation.  Because it includes all of
the works cited by any paper in this volume, it includes many citations
which are not directly related to automatic differentiation.  For
example, it includes basic references in optimization, symbolic algebra
systems, and several applications areas.

    The electronic form of this bibliography contains many other works
relating to automatic differentiation.  The electronic version
is available from netlib ({\tt netlib@research.att. com}).  Corrections and
additions are welcome and should be sent to Dr. George F. Corliss,
Department of Mathematics, Statistics, and Computer Science, Marquette
University, Milwaukee, WI  53233 USA, georgec@boris.mscs.mu.edu.



% These Procedings:
   \nocite{Beda59a,Gilb91a}
   \nocite{Berz91a,Bisc91a,Char91a,Corl91a,Dixo91a}
   \nocite{Evtu91a,Fisc91b,Flan91a,Garc91a,Gayd91a,Gold91a}
   \nocite{Grie91a,Irim91a,Horw91a,Jued91a}
   \nocite{Kalm91a,Kubo91a,Laws91a,Layn91a,Mazo91a,Mich91a}
   \nocite{Navo91a,Rall91a,Saar91a,Shah91a,Soul91a,Tala91a}
   \nocite{Tesf91a,Thac91a,Tham91a,Worl91a}

% Berz:
   \nocite{Brow64a,Berz87b,Berz88a,Berz88b,Berz89c,Berz89a}
   \nocite{Berz90a,Berz90b,Berz90c,Berz90d,Berz90e,Berz90f}
   \nocite{Berz91b,Berz91c,Brow79a,Berz87c,Care87a,Cont80a}
   \nocite{Fore89a,Fore89b,Mats72a,Mats70a,Mats70b,Ritt50a}
   \nocite{Woll65b,Woll65a,Woll67a,Woll68a,Woll87a}

% Bischof:
   \nocite{Bisc91b,Bisc91c,Chri90a,Cole84a,Cole83a,Denn83a,Dixo87a}
   \nocite{Fisc90a,Gold84a,Grie89a,Grie91b,Grie91c,Grie91d,Grie90a}
   \nocite{Jone91a,Jued91a,Jued90a,More90a,More80a,More80b,Navo79a}
   \nocite{Plas90a,Wint79a}

% Char:
   \nocite{Baue74a,Breu69a,Char86a,Cook88a,Dewa89a,Dewa90a,Eise68a}
   \nocite{Engq80a,Fate69a,Free88a,Fitc85a,Gate86a,Gold91a,Grie89a}
   \nocite{Kahr53a,Kalt88a,Kalt90a,Kalt89a,Lana81a,Lisk90a,Linn76a}
   \nocite{Lars78a,Mill75a,Mona91b,Mutr91a,Mazu90a,Nged79a,Nged79b}
   \nocite{Nola53a,Pave85a,Rost91b,Stei86a,Stou77a,Heuv87a,vHul81a}
   \nocite{vHul83a,vHul89a,Wang86a,Wirt81a,Wolf90a}

% Corliss:
   \nocite{Bisc91a,Beck88a,Chan67a,Chan91a,Chan82a,Chan81a,Char88a}
   \nocite{Chri91a,Corl91a,Corl87a,Corl91b,Crar79a,Davi88a,AdaLRM,Fisc91a}
   \nocite{Gray75a,Grie89a,Grie91a,Grie90a,Hill85a,Horw88a,IBM90a}
   \nocite{Irim84a,Irim88a,Klat90a,Moor66a,Moor79a,Rall81a}
   \nocite{Spee80a,Step90a,Stet86a,Stro86a}

% Dixon:
   \nocite{Byrn87a,Chri90a,Dixo87c,Dixo90a,Dixo87d,Dixo89a}
   \nocite{Duff87a,Flet91a,Grie89a,Grie90c,Park90a,Pryc87a}
   \nocite{Rall81a,Tesf91a}

% Evtushenko:
   \nocite{Aida89a,Evtu85a,Evtu89b,Grie89a,Irim84a,Irim87a}

% Fischer:
   \nocite{Fisc91c,Irim84a,Irim87a,Jack86a,Kede80a,McCo83a,Rall81a}

% Flanders:
   \nocite{Berz91a,Flan84a,Henr74a,Laws91a,Neid89a,Nije78a,Rall91a}

% Garcia:
   \nocite{Bard74a,Bigg71a,Bigg73a,Bren73a,Char91a,Davi81a}
   \nocite{Garc79a,Garc83a,Garc84a,Garc88a,Garc88b,Garc89a,Gold91a}
   \nocite{Hill85a,Jued91a,Lill70a,NOC76a,ONei71a,Rall81a,Soul91a}
   \nocite{STSC89a,Turn85a,Weng64a,Zack71a}

% Gay:
   \nocite{Brac68a,Broo88a,Broo83a,Denn81a,Feld90a,Feld83a,Flet65a}
   \nocite{Four90a,Fras91a,Foxp78a,Gayd83a,Grah75a,Grie89a,Grie90a}
   \nocite{Himm72a,More81b,Murt81a,Murt78a,Murt82a,Murt87a,Spee80a}

% Goldman:
   \nocite{Baur83a,Boyl90a,Breu69a,Char86a,Char88a,Dole90a}
   \nocite{Flaj87a,Flaj90a,Gate85a,Gate86a,Gate87a,Gome90a}
   \nocite{Grie89a,Grie90a,Hear87a,Hill85a,Horw88a,vHul81a}
   \nocite{vHul83a,vHul89a,vHul91a,Irim87a,Pave85a,Rall84a}
   \nocite{Wolf88a}

% Griewank:
   \nocite{Cole83a,Dixo91a,Duff86a,Grie89a,Grie90c,Grie90a,Grie91c}
   \nocite{Irim91a,Mill80a,Muro87b,Rose78a,Yosh87a}

% Iri:
   \nocite{Aida89a,Alfe83a,Baue74a,Baur83a,Bord56a,Char91a,Civa65a}
   \nocite{Drey69a,Evtu89a,Evtu91a}
   \nocite{Fisc90b,Grie89a,Grie91a,Hill85a,Horw88a,Irim62a,Jacq65a}
   \nocite{Irim69a,Irim84a,Irim87a,Irim88a,Iwat83a,Jued91a}
   \nocite{Kaba87a,Kaba88a,Kant57a,Kauc84a,Kede80a,Kimk84b}
   \nocite{Kubo89a,Kubo88b,Kubo89b,Linn76a,Maso53a,Mati86a}
   \nocite{Mill80a,Mori90a,Muro87a,Ostr71a,Rall81a,Rote90a,Sawy84a}
   \nocite{Spee80a,Thac88a,Tsuc88a,Ukko81a,Voli85a,Weng64a}

% Horwedel:
   \nocite{Oblo86a,Horw88a,Oblo83a,Worl86a,Irim84a,Thac88a}
   \nocite{Grie89a,Horw89b,Horw90a,Fiel86a,Moor79b}

% Juedes:
   \nocite{Adam69a,Berz87a,Berz89a,Berz90a,Berz90b,Berz90c}
   \nocite{Berz90d,Chan82a,Chri90a,Corl84a,Corl87a,Corl91b}
   \nocite{Corl91a,Flan91b,Four90a,Gayd91b,Garc91a,Grie89a}
   \nocite{Grie90a,Hill85a,Horw88a,Horw89a,Horw90a,Irim87a}
   \nocite{Iwat84a,Jerr89b,Jerr89a,Jerr90a,Jued90a,Krin84a}
   \nocite{Kubo90a,Laws71a,Liep90a,Lind91a,Math89a,Mazo91a}
   \nocite{McCu69a,Mich90a,Mona91a,Neid91a,Pfei87a,PROS77a}
   \nocite{Rall81a,Hill91a,Rost91a,Step91a,Spee80a,Tesf91a}
   \nocite{Tham69a,Tham75a,Tham82a,Tham89a,Tham91a,Yosh89a}

% Kalman:
   \nocite{Elli90a,Pohl89a,Grie91d,Corl91a,Neid89a,Rall84a}
   \nocite{Rall86a}

% Kubota:
   \nocite{Baur83a,Bohl87a,Grie89a,Grie91b,Grie90a,Henr64a}
   \nocite{Hill85a,Irim84a,Irim91a,Irim87a,Irim88a,Iwat84a}
   \nocite{JIS82a,John75a,Jued91a,Kant57a,Kimk84a,Kimk84b}
   \nocite{Kubo88a,Kubo89a,Kubo88b,Kubo89b,Kubo90a,Linn76a}
   \nocite{Muro89a,Rall80a,Spee80a,Ukko81a,Voli85a,Wilk63a}

% Lawson:
   \nocite{Bram61a,Bren78a,Bren77a,Knut81a,Laws71a,Math89a}
   \nocite{Laws88a,Nije78a,Thac66a,Weng64a}

% Layne:
   \nocite{Abra70a,Bisc91a,Bohl87a,Corl88a,Corl91a,Esco76a}
   \nocite{Fisc91a,Kuli86a,Lohn87a,Moor79a,Mort74a,Pres86a}
   \nocite{Rall81a,Rall83a,Skee89a,Stet90a,Wert88a}

% Mazourik:
   \nocite{Rall81a,Grie89a,Jued91a}

% Michelotti:
   \nocite{Ande79a,Berz89b,Berz87b,Fore89a,Grie90a,Kala83a}
   \nocite{Kede80a,Mich90a,Rall86a}

% Navon:
   \nocite{Marc74a,Pene76a,LeDi82a,Cour85a,Tala87a,Lewi85a}
   \nocite{Derb85a,LeDi86b,Cour87a,Sela80a,Navo90a,Grie89a}
   \nocite{Noce80a,Liud88a}

% Rall:
   \nocite{Corl87a,Corl91b,Grie89a,Henr74a,Irim84a,Moor79a}
   \nocite{Rall79a,Rall81a,Rall83a,Rall83b,Rall90a,Rats84a,Spee80a}
 
% Saarinen:
   \nocite{Arbi87a,Duff74a,Grie89a,Hert91a,McCu43a,Mins88a}
   \nocite{Rose63a,Rume86a,Saar91b,UNIC91a,Werb74a,Werb90a}
   \nocite{Werb91a}

% Shah:
   \nocite{Udwa76a,Shah76a,Chen74a,Jacq65a,Chav75a,Cart74a}
   \nocite{Peac55a,Isaa66a}

% Soulie:
   \nocite{Soul88a,Grie89a,AFNO83a,More81a,Ball73a,Ball69a}
   \nocite{Hill85a,Garc91a,Soul86a,Hein71a,Horw88a,Soul89a}
   \nocite{Kubo90a,Step90a,Yosh89a}

% Talagrand:
   \nocite{Morg84a,Lion71a,Marc74a,Hall82a,Cacu84a,Lore81a,Pene76a}
   \nocite{Lewi85a,LeDi86a,Lore88a,Cour90a,Thac88a,Cour90b,Navo90b}
   \nocite{Chao90a,Doua90a,Cour87b,Parl80a,Urba85a,Veyr90a,LaCa88a}
   \nocite{Thac91a}

% Tesfatsion:
   \nocite{Bell65a,Datt91a,Grie89a,Grie90a,Haye87a,Hort89a,Huss90a,Jerr89a}
   \nocite{Kagi85a,Kala81a,Kala86a,Kala90a,Kala91a,Kala81b,Kala83c,Kala87a}
   \nocite{Kala84a,Kede80a,Kill85a,Neid90a,Pfei80a,Rall81a,Rall84a,Rall86a}
   \nocite{Spee80a,Voli85a,Wang84a,Weng64a,Wexl87a,Wexl88a}

% Thacker:
  \nocite{Ahoa86a,Grie89a,Irim84a,Spee80a,LeDi86a,Thac90a,Thac88a}

% Thames:
  \nocite{Adam69a,Bell65b,Krin84a,McCu69a,Pfei87a,PROS77a,Tham69a}
  \nocite{Tham75a,Tham82a,Tham89a}

% Worley:
   \nocite{Tomo72a,Stac74a,Gree76a,Oblo78a,Fran78a,Cacu81a,Cacu81b}
   \nocite{Weis82a,Radi91a,Jued91a}


\bibliography {all_brec}
\bibliographystyle{brecsiam}

\end {document}


.
